Author's response to reviews

Title: Effects of eccentric exercise in patients with subacromial impingement syndrome: a systematic review and meta-analysis

Authors:

Robin Larsson (ljrobin.larsson@gmail.com)

Susanne Bernhardsson (susanne.bernhardsson@vgregion.se)

Lena Nordeman (lena.nordeman@vgregion.se)

Version: 1 Date: 16 Jul 2019

Author’s response to reviews:

1. PRISMA Guidelines - done
2. Title page - done
3. Abbreviations - done
4. Figures - done